Author's response to reviews

Title: Circulating endothelial cells are an early predictor in renal cell carcinoma for tumor response to sunitinib

Authors:

Viktor Grünwald (Gruenwald.Viktor@MH-Hannover.de)
Gernot Beutel (Beutel.Gernot@MH-Hannover.de)
Susanne Schuch-Jantsch (JantschSusanne@web.de)
Christoph Reuter (Reuter.Christoph@MH-Hannover.de)
Philipp Ivanyi (Ivanyi.Philipp@MH-Hannover.de)
Arnold Ganser (Ganser.Arnold@MH-Hannover.de)
Marion Haubitz (Haubitz.Marion@MH-Hannover.de)

Version: 7 Date: 13 December 2010

Author's response to reviews:

revision were performed according to suggested changes.
Titel and abstract have been updated.
Authors list and references changed accordingly.
Tracking has been removed.
Competing interest and author contribution has been re-formatted.
vertical lines of tables were removed.